Author's response to reviews

Title: Massive parallel sequencing as a new diagnostic approach for phenylketonuria and tetrahydrobiopterin-deficiency in Thailand

Authors:

Pongsathorn Chaiyasap (chaiyasap.p@gmail.com)
Chupong Ittiwut (chupongi@gmail.com)
Chalurmon Srichomthong (chalurmon_s@hotmail.com)
Apiruk Sangsin (oak7148@hotmail.com)
Kanya Suphapeetiporn (kanya.su@chula.ac.th)
Vorasuk Shotelersuk (vorasuk.s@chula.ac.th)

Version: 3 Date: 02 Sep 2017

Author’s response to reviews:

Editor
BMC Medical Genetics

September 2, 2017

Dear Editor,

Manuscript ID MGTC-D-17-00124R2 entitled “Massive parallel sequencing as a new diagnostic approach for phenylketonuria and tetrahydrobiopterin-deficiency in Thailand”

Thank you very much for carefully reviewing our manuscript and giving us very useful comments. We have made changes in the manuscript according to the reviewer/editor’s comments or suggestions as the followings:

Editor Comments:

Q1. As part of our editorial policies we strongly encourage all authors to deposit their sequencing data before publication. Please deposit the sequencing data generated in this study in a suitable public repository such as the NCBI SRA database. Once you have deposited your data, ensure you provide the information on deposition and how to access these data in the ‘Availability of
Data and Materials’ section in Declarations, including the permanent link or the unique identifier associated to it.

A1: As kindly suggested, the sequencing data generated in this study have been deposited in the public repository, the NCBI SRA database. The relevant information and the permanent link have been added in the ‘Availability of Data and Materials’ section. Please see the revised manuscript on page 10, lines 7-12.

Q2. Please ensure that all of the authors’ names are entered into Editorial Manager correctly, two of the authors appear to have their surname entered twice. If this is the way their names are meant to be spelled, please ignore this comment.

A2: Thanks to the editor, all the authors’ names have been checked and entered into Editorial Manager correctly.

Q3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

A3: As kindly suggested, the manuscript has been uploaded as a single, final, and clean version.

Thank you very much. We really appreciate all your kind assistance. We hope that it meets all the requirements of the journal.

Sincerely,

Kanya Suphapeetiporn, MD, PhD
Corresponding Author